

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Zuchowski, David

## Working Paper Migration response to an immigration shock: Evidence from Russia's aggression against Ukraine

Ruhr Economic Papers, No. 1039

**Provided in Cooperation with:** RWI – Leibniz-Institut für Wirtschaftsforschung, Essen

*Suggested Citation:* Zuchowski, David (2023) : Migration response to an immigration shock: Evidence from Russia's aggression against Ukraine, Ruhr Economic Papers, No. 1039, ISBN 978-3-96973-208-3, RWI - Leibniz-Institut für Wirtschaftsforschung, Essen, https://doi.org/10.4419/86788924

This Version is available at: https://hdl.handle.net/10419/281188

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# RUHR ECONOMIC PAPERS

David Zuchowski

Migration Response to an Immigration Shock: Evidence from Russia's Aggression against Ukraine





#1039

## Imprint

#### Ruhr Economic Papers

#### Published by

RWI – Leibniz-Institut für Wirtschaftsforschung Hohenzollernstr. 1-3, 45128 Essen, Germany Ruhr-Universität Bochum (RUB), Department of Economics Universitätsstr. 150, 44801 Bochum, Germany Technische Universität Dortmund, Department of Economic and Social Sciences Vogelpothsweg 87, 44227 Dortmund, Germany Universität Duisburg-Essen, Department of Economics Universitätsstr. 12, 45117 Essen, Germany

#### Editors

## Prof. Dr. Thomas K. Bauer RUB, Department of Economics, Empirical Economics Phone: +49 (0) 234/3 22 83 41, e-mail: thomas.bauer@rub.de Prof. Dr. Ludger Linnemann Technische Universität Dortmund, Department of Business and Economics Economics – Applied Economics Phone: +49 (0) 231/7 55-3102, e-mail: : Ludger.Linnemann@tu-dortmund.de Prof. Dr. Volker Clausen University of Duisburg-Essen, Department of Economics International Economics Phone: +49 (0) 201/1 83-3655, e-mail: vclausen@vwl.uni-due.de Prof. Dr. Ronald Bachmann, Prof. Dr. Manuel Frondel, Prof. Dr. Torsten Schmidt, Prof. Dr. Ansgar Wübker RWI, Phone: +49 (0) 201/81 49-213, e-mail: presse@rwi-essen.de

#### **Editorial Office**

Sabine Weiler

RWI, Phone: +49 (0) 201/81 49-213, e-mail: sabine.weiler@rwi-essen.de

#### Ruhr Economic Papers #1039

Responsible Editor: Ronald Bachmann

All rights reserved. Essen, Germany, 2023

ISSN 1864-4872 (online) - ISBN 978-3-96973-208-3

The working papers published in the series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editors.

Ruhr Economic Papers #1039

David Zuchowski

Migration Response to an Immigration Shock: Evidence from Russia's Aggression against Ukraine



## Bibliografische Informationen der Deutschen Nationalbibliothek

The Deutsche Nationalbibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data are available on the Internet at http://dnb.dnb.de

RWI is funded by the Federal Government and the federal state of North Rhine-Westphalia.

http://dx.doi.org/10.4419/96973208 ISSN 1864-4872 (online) ISBN 978-3-96973-208-3 David Zuchowski<sup>1</sup>

# Migration Response to an Immigration Shock: Evidence from Russia's Aggression against Ukraine

## Abstract

Russia's attacks against Ukraine have triggered massive and unexpected migration movements. In this paper, I examine the impact of the inflow of Ukrainians that resulted from Russia's aggression in 2014 on local migration patterns in Poland. For identification, I use an instrumental variable approach drawing on unique historical data on the forced resettlement of Ukrainians in Poland after World War II. The results show that the regional inflow of immigrants decreases both internal and international out-migration of the Polish population. I provide supportive evidence that the decrease in out-migration is due to the upscaling of local labor markets.

JEL-Codes: F22, J61, O15

Keywords: Migration; immigrant workers; Poland; Ukraine

January 2024

<sup>1</sup> David Zuchowski, RWI and UDE. - I am grateful to Günes Asik, Ronald Bachmann, Julia Bredtmann, Gökay Demir, Anthony Edo, Ismael Gálvez Iniesta, Yvonne Giesing, Roman Klauser, Fernanda Martínez Flores, Dariia Mykhailyshyna, Sebastian Otten, João Pereira dos Santos, Eduard Storm, Oliver Wach, Giulio Zanella, two anonymous reviewers, and participants of the BeNA Winter Workshop 2022 in Berlin, the 4th EBRD-King's College London Workshop on the Economics and Politics of Migration in Istanbul, the XXV Applied Economics Meeting in Toledo, the Junior Workshop on the Economics of Migration in Paris, the Workshop on the Economics of Firms and Labor 2023 in Utting am Ammersee, the Brown Bag at the University of Duisburg-Essen, and RWI internal seminars for helpful comments and suggestions. Rachel Kühn and Jurek Tiedemann provided excellent research assistance. I declare to have no relevant or material financial interests that relate to the research described in this paper. - All correspondence to: David Zuchowski, RWI, Hohenzollernstr. 1-3, 45128 Essen, Germany, e-mail: david.zuchowski@nvi-essen.de

## 1 Introduction

Massive and unexpected migration flows emerged due to Russia's attacks against Ukraine. The migration surge began with the annexation of Crimea in 2014 and reached unprecedented dimensions after the full invasion of Ukraine in February 2022, when more than 8 million people fled to neighboring countries (UNHCR, 2023). Central and Eastern European countries such as Estonia, the Czech Republic, Poland, and Lithuania are the largest per capita recipients of Ukrainian refugees among OECD countries (OECD, 2022a).<sup>1</sup> Those post-communist countries in transition are not typical immigration countries. On the contrary, since the fall of the Iron Curtain, they have experienced long-standing high emigration, mainly of young and well-educated citizens who saw no professional prospects in their own country (IMF, 2014, 2016).

In this paper, I examine the impact of the inflow of Ukrainians that resulted from Russia's aggression in 2014 on local migration patterns in Poland. It is the first study evaluating the impact of an immigration shock on local migration patterns in a postcommunist country in transition. In particular, I take advantage of the fact that Poland, the country currently hosting the largest absolute number of Ukrainian refugees (UNHCR, 2023), has been the primary destination of conflict-induced temporary labor migration from Ukraine for nearly a decade. The annexation of Crimea by Russia in 2014 and the ongoing conflict in the Donbas region destabilized the Ukrainian economy, forcing many Ukrainians to look for work abroad. As a result, the Polish labor market has experienced a massive and unexpected conflict-induced labor supply shock.

The analysis in this paper focuses on migration outcomes, which are particularly interesting as Poland, like other post-communist countries, has for decades been a country of high labor emigration, especially of well-educated citizens. The education boom during the socio-economic transition and the lack of demand for highly educated workers led to graduates of Polish universities being employed as "cleaners in Europe, dishwashers in

<sup>&</sup>lt;sup>1</sup>See also Figure A1 in the Appendix. Overall, the country with the highest per capita inflow of Ukrainian refugees is Moldova (United Nations, 2022), the last European country to overthrow the communist government (Barsbai *et al.*, 2017).

London, and caregivers for the elderly in Germany" (Radio Poznan, 2011). Furthermore, with around 2 million temporary foreign workers in 2019, the vast majority of whom were Ukrainians, Poland underwent a rapid transformation from a traditional emigration country to a host country (Statistics Poland, 2020). This sudden and massive inflow of temporary workers could potentially further increase Polish emigration, for example, by intensifying competition in local labor markets. However, with the inflow of Ukrainian workers after the Russian aggression in 2014, Polish emigration decreased (Statistics Poland, 2020), and at the same time the share of highly skilled emigrants among all Polish emigrants also decreased (Giesing and Schikora, 2023).

To examine the impact of the inflow of Ukrainian workers caused by Russia's aggression in 2014 on the migration behavior of the local population, this paper exploits the spatial variation across Polish counties in the intensity of exposure to this labor supply shock. I use administrative data on internal and international migration to measure changes in migration patterns in Poland and administrative records on firms' statements on the employment of foreigners to measure the intensity of the local exposure to the labor supply shock.

The main challenge is the endogeneity of immigrant workers' location decisions. Despite the labor supply shock's exogenous timing, the location of Ukrainian workers across Polish counties is, at least to some extent, determined by the prospects of the local labor markets, which are also determinants of changes in the migration behavior of natives (Peri, 2016). Thus, counties attracting more immigrant workers could have experienced higher in-migration trends even in the absence of the inflow of Ukrainian workers.

To address this issue, I propose a novel instrument based on unique historical records on the forced resettlement of Ukrainians in Poland in the aftermath of World War II. This mass displacement took place as part of a military operation called Vistula, during which around 140,000 Ukrainians were forcibly relocated. The instrument draws on the commonly used distance instrument. However, instead of instrumenting the contemporary distribution of Ukrainian workers with the distance to the border of immigrants' country of origin, I use the distance to historical hotspots of Ukrainian networks in Poland that emerged due to Operation Vistula.<sup>2</sup> First, I define the historical hotspots as places where the share of forcibly resettled Ukrainians exceeded 10% of the local population in 1950.<sup>3</sup> Second, I instrument the contemporary location of Ukrainians with the distance in kilometers to the nearest hotspot point. The constructed distance instrument is predictive of the contemporary location of Ukrainians and thus relevant. Furthermore, it does not predict changes in migration patterns or any other socio-economic characteristics in the pre-treatment period, which suggests that the exclusion restriction is satisfied.

The findings show that the immigrant inflow decreases both internal and international out-migration. On average, the inflow of 1000 Ukrainian workers into local labor markets lowers internal out-migration by around 19 inhabitants and international out-migration by about 5 inhabitants. I find evidence of upscaling, that is, an enlargement of local labor markets as a potential driver of the results. As descriptive evidence suggests complementarity between Ukrainian workers and Polish emigrants, this may indicate the absorption of local workers into the local labor market who would otherwise emigrate. Furthermore, I find evidence of crowding-out in-migration as an initial response to the shock. This effect decreases in magnitude and becomes statistically insignificant for international in-migration after two and for internal in-migration after four years, showing a reduction of this effect in the long run.

I run several robustness checks. The results are robust to a number of different specifications and the exclusion of potential outliers. Furthermore, I find no evidence of pre-treatment trends in migration outcomes between counties with different exposure to the inflow of Ukrainians, which suggests that the parallel trend assumption holds.

The findings have important policy implications, especially in light of the escalation of Russian aggression against Ukraine in 2022 and the subsequent decision to invoke the Temporary Protection Directive in the European Union. The latter provided Ukrainian refugees unprecedented free access to host countries' labor markets and thus led to a significant labor supply in several post-communist countries. Estimates suggest that the

<sup>&</sup>lt;sup>2</sup>For example, Dustmann *et al.* (2016) and Aksu *et al.* (2022) use the distance to the border of immigrants' country of origin as an instrument in their studies on the effects of immigration.

<sup>&</sup>lt;sup>3</sup>The results are robust to alternative definitions of the threshold.

inflow of Ukrainian refugees increases the labor force in the Czech Republic, Poland, and Estonia by around 2%, and many other countries in Central and Eastern Europe by 1 to 1.5% (OECD, 2022b). This sudden labor supply shock could pose a particular challenge for post-communist countries, which, despite their transition to capitalist economies, still have relatively weak labor market institutions and no established immigration structures. However, the present study finds that in transition economies with a surplus of highly educated workers, an immigration shock can result in positive effects for host communities.

This article relates to the broad debate on the impact of immigration on host counties' labor markets, in particular, to the growing literature analyzing the impact of immigration on the migration behavior of the local population.<sup>4</sup> Several recent studies provide evidence on the effects of conflict-induced immigration and find mixed results. Batut and Schneider-Strawczynski (2022) find a decline in the in-migration to municipalities hosting refugees in France. Elmallakh and Wahba (2023) examine the impact of the inflow of Syrian refugees on Jordanians' migration behavior and find an increase in internal out-migration from affected regions but an increase in job-related in-migration into the camp areas. On the contrary, in a similar analysis for Turkey, Akgündüz *et al.* (2021) find a decrease in inter-regional job-related in-migration to provinces hosting Syrian refugees but no evidence of an increase in out-migration. However, in the context of conflict-induced internal displacement in Colombia, Morales (2018) finds an increase in out-migration from affected communities, in particular of high-skilled workers. Moreover, empirical findings in the setting of immigration to highly developed countries are also not homogenous and indicate that the effects depend on the immigration policy and the type of migration.<sup>5</sup>

This paper contributes to the literature by providing evidence from a unique and highly policy-relevant context. As forced migrants typically seek protection either in

<sup>&</sup>lt;sup>4</sup>For a broad literature survey on the impact of immigration in OECD countries see for example Edo *et al.* (2020). Furthermore, Verme and Schuettler (2021) provide a recent review on the impact of forced displacement on host countries. Most literature studying the impact of immigration on host countries has focused on the impact of immigration on wages and employment. See Edo (2019) or Dustmann *et al.* (2016) for an extensive literature review.

<sup>&</sup>lt;sup>5</sup>See for example Card (2001) and Borjas (2006) for evidence for the US, Hatton and Tani (2005) for UK, Mocetti and Porello (2010) for Italy, Beine and Coulombe (2018) for Canada, Moraga *et al.* (2019) for Spain, Ortega and Verdugo (2022) for France, and Han *et al.* (2022) for South Korea.

neighboring developing countries or in high-income OECD countries (Devictor *et al.*, 2021), little is known about the impact of conflict-related immigration in other settings. The case of Poland is also unique in that it is a post-communist country with long-standing high emigration. Moreover, at the time of the immigration shock, Polish citizens enjoyed the freedom of migration within the European Union (including labor migration), which allowed for a potentially strong migration response. Finally, unlike most previous studies, my data allows me to examine both the internal and the international migration response of the local population.

The remainder of this paper is organized as follows. Section 2 provides background information on the inflow of Ukrainians to Poland. Section 3 describes the underlying data. Section 4 discusses the empirical strategy. Section 5 presents and discusses the empirical results. Section 6 concludes.

## 2 Background Information

## 2.1 Destabilization of Ukraine and Labor Migration to Poland

After the collapse of the Soviet Union in 1991, Ukraine became an independent state in its current borders. The newly established Ukraine was divided on whether it should be more economically and politically aligned with the West or with Russia. Two decades later, however, when former Ukrainian President Viktor Yanukovych refused to sign the European Union - Ukraine Association Agreement, massive protests emerged on Kyiv's Independence Square. The protests from November 2013 to February 2014, later called Euromaidan, became a symbol of Ukrainians' support for closer relations with the West. Following the escalation of protests, the pro-Russian President Yanukovych fled Kyiv and later Ukraine (Plokhy, 2016).

Taking advantage of the internal crisis in Ukraine, Russia took control of the Crimean Peninsula. Furthermore, the subsequent fights and the emergence of pro-Russian selfproclaimed "people's republics" in the eastern Ukrainian territories of Donetsk and Lugansk further intensified the political and economic destabilization of Ukraine (Plokhy, 2016).

The destabilization of the Ukrainian economy following the Russian aggression has forced many Ukrainians to search for work abroad. The solid line in Figure 1a shows a sharp depreciation of Ukraine's national currency after February 2014. This sharp fall in the exchange rate of the Ukrainian hryvnia against the Polish zloty increased the economic incentives for Ukrainians to work in Poland, as the depreciation amplified the purchasing power of the earnings in Polish zloty when converted to Ukrainian hryvnia. The devaluation of the Ukrainian currency aligns with the increase in interest in labor migration from Ukraine to Poland.<sup>6</sup> This trend is illustrated by the dashed line, which shows the sudden rise in Google queries in Ukraine for "work in Poland" after the Russian aggression.

Alongside the income prospects, the existence of networks in the destination country



Figure 1: INFLOW OF UKRAINIANS TO THE POLISH LABOR MARKET

*Notes:* Panel (a): Polish zloty to Ukrainian hryvnia exchange rate based on official data from the National Bank of Poland and relative intensity of google searches in Ukraine for "work in Poland" (in Ukrainian) provided by Google Trends. Panel (b): Number of firms' statements on the employment of a foreigner over time. In both panels, the vertical gray lines indicate the period before and after the Russian aggression against Ukraine.

<sup>&</sup>lt;sup>6</sup>According to Statistics Ukraine (2017), between 2015 and 2017, over 73% of Ukrainian labor migrants in Poland originated from Western Ukraine, not directly affected by the war. This suggests that the migration shock was primarily driven by the economic and currency shock resulting from Russian aggression.

is a driver of migration (Docquier *et al.*, 2014). Ukrainian networks in Poland as well as relatively small geographical and linguistic distance, which also play a role in the choice of destination (see e.g., Adsera and Pytlikova (2015), and Bredtmann *et al.* (2020)), make Poland a natural destination country for Ukrainians.<sup>7</sup> Thus, after the Russian aggression against Ukraine in 2014, the Polish economy experienced an unexpected and massive inflow of Ukrainian workers reaching almost two million in 2019 as illustrated in Figure 1b. The next section discusses their geographic location within Poland.

# 2.2 Ukrainians in Poland: Historical Background and Geographic Dispersion

Figure 2 shows the geographic dispersion of Ukrainian workers across Polish counties in 2019. Ukrainians are more likely to work in areas of Poland far from the Ukrainian border for historical reasons.



Figure 2: Geographic dispersion of Ukrainian workers in 2019

*Notes:* This figure presents the spatial distribution of the number of statements on the employment of a Ukrainian in 2019 (proxy for the number of Ukrainian workers) standardized by the pre-shock working-age local population in 2013 across Polish counties.

<sup>&</sup>lt;sup>7</sup>For literature on the role of migrant networks in determining migration and location decisions, see for example McKenzie and Rapoport (2010), Beine and Coulombe (2018), and Giulietti *et al.* (2018).

After World War II, most Ukrainians living within Poland's new borders were resettled to the Soviet Union. The remaining Ukrainian minority, living primarily in the southeast of the Polish territory, was forcibly resettled by the Polish People's Army in April 1947. Overall, around 140,000 Ukrainians were relocated during the military operation called Vistula. The official goal of this forced resettlement was to prevent the recruitment of new members of the Ukrainian Insurgent Army (UPA) and suppress any potential collaboration with this Ukrainian nationalist paramilitary formation (Nasz Wybir, 2016; Plokhy, 2016; Misiło, 2013).

During the military operation Vistula, Ukrainians living in southeastern Poland were to be "scattered among the Polish population so that they do not pose any danger [to the Polish People's Republic]" (Misiło, 2013). Entire families identified by the local government as Ukrainian, along with mixed families, were forced to leave the areas they had historically inhabited. To undermine the national, cultural, and familial ties within the Ukrainian community in Poland, people originating from the same locality were supposed to be separated, and the total number of resettled Ukrainians was not to exceed 10% of the local population. These guidelines often went unmet due to two primary reasons: delayed arrival of information to the authorities or receiving it only after resettlement, as well as the initial underestimation of the Ukrainian population's size, resulting in a much larger resettlement than expected. As a result, the allocation of Ukrainians to counties was influenced by the extent of post-war destruction and thus by the availability of housing. The crucial factor, however, was the availability of train stations, since the resettlement was primarily carried out using trains (Misiło, 2013).

Furthermore, Polish authorities' efforts to create a unitary state and suppress Ukrainian national identity led to the prohibition of the Ukrainian language and culture in the early 1950s (Nasz Wybir, 2016). The dispersion of the Ukrainian population and further persecution made it difficult to preserve the Ukrainian language and cultivate culture. Therefore, only if the resettled Ukrainian community was large enough to resist the forced cultural assimilation, the Ukrainian identity could survive. As restrictions on the cultivation of Ukrainian culture were gradually eased over time, Ukrainian institutions emerged in large Ukrainian centers, that had formed as a consequence of Operation Vistula. Establishments such as Ukrainian churches or schools helped to preserve Ukrainian culture and language despite earlier dispersion and repression.<sup>8</sup> The existence of such Ukrainian establishments and networks that attracted contemporary migrant workers from Ukraine explains their current distribution far from the Ukrainian border in places not populated by Ukrainians before World War II.

## 2.3 Complementarity of Polish Emigrants and Foreign Workers

Well-established immigration countries, as the United States, Canada, or Germany, seek to alleviate labor shortages through immigration. In contrast, before the inflow of temporary workers from Ukraine, Poland was a country of emigration. Over 2 million Poles emigrated from Poland in the decade after the country's accession to the European Union in 2004 (Financial Times, 2014). Similar to trends in other post-communist countries, Polish emigrants were on average younger and more educated than the overall population. In fact, the proportion of Polish emigrants with tertiary education was approximately twice as high as that within the Polish population (IMF, 2016).

The positive selection of educated emigrants is related to the saturation of the Polish labor market with university graduates. After the fall of the Iron Curtain, Poland began a transition from a communist to a free-market country, which led to a surge in demand for university education. The desire for better living conditions and a social advancement has led to a massive increase in the number of people with higher education (Onet, 2012a). Figure 3a shows that the share of young people with tertiary education in Poland increased from around 10-15% in the 1990s to almost 45% in 2014.

<sup>&</sup>lt;sup>8</sup>Ukrainian religious ceremonies were allowed to be cultivated, first at Polish Roman Catholic parishes and later also in newly established Ukrainian Greek Catholic and Orthodox churches. See Figure A2 in the Appendix for examples. Further liberties were introduced after social protests in the 1980s and the subsequent collapse of communism in Poland (Nasz Wybir, 2016). For example, schools with education in the Ukrainian language were established in the 1990s. Figure A3 in the Appendix shows two examples of such schools.



**Figure 3:** COMPLEMENTARITY OF POLES ENTERING LABOR MARKET AND UKRAINIAN IMMIGRANTS *Notes:* Panel (a): Percentage of 25 to 34 years old with tertiary education over time. Data source:

OECD (2023) The vertical gray line indicates the periods before and after the treatment. Panel (b): Distribution of the share of statements on the employment of a Ukrainian in 2019 by occupation groups according to the Polish Classification of Occupations and Specializations.

Despite experiencing steady economic growth, the Polish economy was not able to absorb the excessive number of young people with higher education entering the Polish labor market "saturated with freshly baked managers, teachers, economists, lawyers, sociologists, and political scientists" (Dziennik Gazeta Prawna, 2012a). The record-breaking unemployment rates of people with higher education (Dziennik Gazeta Prawna, 2012b), lack of job prospects in the profession after graduation, and the persistence of relatively low wages in Poland led to the "exodus of youth" (Financial Times, 2014).

Given Poland's long-standing high emigration rates, the massive and sudden inflow of Ukrainian workers could increase the Polish '*exodus*', for example, by intensifying competition in the affected local labor markets. However, Polish emigration decreased in parallel with the inflow of temporary workers from Ukraine.<sup>9</sup> Furthermore, the share of highly educated emigrants in all Polish emigrants has decreased from a long-standing trend of above 40% to below 35% (Giesing and Schikora, 2023).<sup>10</sup> Could the unexpected

<sup>&</sup>lt;sup>9</sup>See Figure A4 in the Appendix.

<sup>&</sup>lt;sup>10</sup>This indicates a decrease in brain drain from Poland, as the overall share of tertiary education attainment in Poland remained stable after the inflow of temporary workers from Ukraine. In particular, the educational attainment of the 25-34 age group, which was most likely to emigrate (Giesing and Schikora, 2023), also remained stable (see Figure 3a).

inflow of temporary workers from Ukraine be contributing to the drop in Polish emigration, particularly of highly educated citizens?

Whereas Polish graduates often preferred to emigrate and take up manual work abroad to avoid low-paying positions below their qualifications in Poland (Onet, 2012b), Ukrainians were willing to accept even low-wage jobs in Poland. As temporary foreign workers spend a significant share of their income in the home country, their reservation wage is determined by the relative exchange rate between the currencies of their home and host country (Dustmann *et al.*, 2021). Figure 1a shows that the value of the Polish zloty increased from roughly 2 to around 7 hryvnia after Russian aggression. Thus, whereas Ukrainian employees were not necessarily low-skilled, they were likely to have a lower reservation wage. As a result, Ukrainian workers were willing to do even low-wage jobs that natives were not. Furthermore, Figure 3b shows that almost 90% of Ukrainian workers were thus engaged in occupations complementary to the skills of the highly educated Polish emigrants.

## 3 Data

To estimate the impact of the inflow of Ukrainian workers on the local migration patterns in Poland, I exploit panel data at the county level.<sup>11</sup> I combine multiple administrative data and unique historical data based on military records.

*Migration data.* I use two sources of administrative data on migration. First, data on the internal and international migration patterns, my main outcome variables, is provided by the Central Statistical Office of Poland (Statistics Poland). This county-level data is based on administrative records from the Polish Universal Electronic Population Registration System, which centrally collects from local authorities all information on

<sup>&</sup>lt;sup>11</sup>Counties are the second administrative level (NUTS-4) in Poland. My original sample includes all 380 counties. Due to administrative changes, I aggregate the data of Walbrzych county and the city with county rights Walbrzych, which was part of Walbrzych county until 2013. This results in a final sample size of 379. Table A1 in the Appendix reports the descriptive statistics.

mandatory (de-)registrations for a permanent stay.<sup>12</sup>

Second, to measure the spatial distribution of Ukrainian workers across Polish local labor markets, I use county-level administrative records of the Ministry of Family, Labor and Social Policy on firms' statements on the employment of a foreigner.<sup>13</sup> In particular, I use variation across Polish counties in the firms' statements on hiring Ukrainian citizens in 2019 as a proxy for the exposure intensity to the labor supply shock. This has twofold reasons. First, this data is not available at the county level before 2019. Second, to carry out the baseline analysis without the migration activity distortion due to the COVID-19 pandemic, I restrict the analysis to the years before 2020.

*Historical data for the instrument.* To generate the instrument, I take advantage of unique historical data. In particular, I construct a novel data set with distances from places to which Ukrainians were forcibly resettled in 1947 during the military operation Vistula described in section 2.2. First, I geolocate data based on archive military records collected by Misiło (2013). Second, to calculate the relative size of the historical network of Ukrainians in Poland, I combine the available information on the size of the military transports with data from Becker *et al.* (2020) on the population size of affected counties based on the first census in Poland after World War II from 1950.

*Mechanisms and control variables.* My primary source of administrative data on mechanisms and control variables at the county level is Statistics Poland, which processes and provides administrative data, including contemporary local labor market information, collected from local authorities. In addition, for historical controls such as the number of railway stations in 1964, the measure of World War II destruction, and industrial production per capita in 1954, I draw on data from Becker *et al.* (2020).

<sup>&</sup>lt;sup>12</sup>Therefore, this data does not encompass the inflow of Ukrainians, as they were only granted a temporary stay in Poland and thus were not eligible for permanent residency registration.

<sup>&</sup>lt;sup>13</sup>Until 29 January 2022, citizens of Ukraine, Armenia, Belarus, Georgia, Moldova, and Russia were allowed to work in Poland without a work permit during a period not exceeding 6 months in consecutive 12 months on the basis of a firm's statement of the intention to employ a foreigner.

## 4 Empirical Strategy

#### 4.1 Econometric Equation

I examine the impact of the inflow of Ukrainian temporary workers on the migration behavior of Poles by estimating the following baseline equation in first differences:

$$\Delta y_c = \beta \Delta m_c + \tau + \Delta \varepsilon_c. \tag{1}$$

Following Edo (2020) and Bohnet *et al.* (2022), I measure the intensity of the labor supply shock as  $\Delta m_c = M_{c,2019}/L_{c,2013}$  where  $M_{c,2019}$  is a proxy for the number of Ukrainian workers in county *c* standardized by the pre-shock working-age local population  $L_{c,2013}$ .  $\Delta y_c = (Y_{c,2019} - Y_{c,2013})/L_{c,2013}$  is the change in either internal or international migration patterns, i.e. in-migration or out-migration, between 2019 (post-shock period) and 2013 (pre-shock period) in county *c*, standardized by the pre-shock working-age local population  $L_{c,2013}$ .  $\tau$  represents the common time trend and  $\Delta \varepsilon_c$  is the error term.

The dependent variable and the explanatory variables of interest are standardized by the same denominator. This allows me to account for the differential sizes of the local labor markets and interpret the coefficient of interest  $\beta$  as the absolute change in the number of migrants associated with the inflow of 100 foreign workers (Han *et al.*, 2022).<sup>14</sup>

First differences eliminate time-invariant county-specific characteristics and  $\tau$  controls for the common time trend of the dependent variable and shocks that affect all counties in the same manner. Furthermore, the first difference specification with two time periods that includes the constant  $\tau$  also accounts for potential structural data misreporting that is constant within counties over time or that changes over time but is constant across all counties. Furthermore, I use heteroskedasticity-robust standard errors and cluster them at the NUTS-3 level (one level higher than the county level) to account for possible serial correlation in migration patterns at the regional level.

The estimation of the first differences model from equation 1 is equivalent to a difference-

<sup>&</sup>lt;sup>14</sup>For visualization purposes, in contrast to Han *et al.* (2022), I multiply the outcome variables by 100.

in-differences model with variable treatment intensity (Edo, 2020). Thus, my identification strategy relies on the common trend assumption. To test the validity of this assumption and examine the dynamics of the effect, I obtain the event study coefficients by interacting the instrumented inflow of Ukrainians in 2019 with yearly indicators and defining  $\Delta m_c$  as the difference in internal or international migration patterns between the respective postor pretreatment year and 2013.

## 4.2 Endogeneity of Immigrants' Location Choices

The identifying assumption of equation 1 is that county-specific time-varying differences included in  $\Delta \varepsilon_c$  are not related to the spatial location of Ukrainian workers. Yet, although the inflow of Ukrainians was unexpected, the location of Ukrainian workers across local labor markets in Poland is not random. Migrants' location choice is partly determined by the prospects of local labor markets, which are also determinants of the migration behavior of natives (Peri, 2016). Thus, counties attracting more immigrant workers would potentially have experienced higher in-migration trends even in the absence of the inflow of Ukrainian workers.

Figure 4 presents the results of a balancing test. It shows that several county characteristics, which potentially have an impact on migration patterns, are highly correlated with the location of Ukrainian workers. In particular, Ukrainians are more likely to choose counties with initially higher wages and lower unemployment. Therefore, a simple OLS model will most likely capture a spurious correlation between the migration patterns of Poles and the presence of Ukrainians driven by the prospects of local labor markets.

To address the endogenous location of immigrants, I make use of the instrumental variable approach. However, conventional instruments used in migration literature, such as shift-share or distance to the immigrants' country of origin, are not an adequate source of exogenous variation in the context of this study. As a result of the forced resettlement of Ukrainians shortly after World War II, Ukrainian networks in Poland do not follow the pattern typical to bordering countries. Furthermore, due to the subsequent persecution of



Notes: This balancing test presents bivariate correlations with the share of Ukrainians (Standardized Beta Coefficients +95% CI). Due to data availability, I use the change in the share of graduates between 2012 and 2013 instead of the change between 2007 and 2013.

Ukrainian culture and language, in regions with a small resettled Ukrainian population, Ukrainians became assimilated. As a result, the historical share of Ukrainians is not predictive of the contemporary location choices of Ukrainian workers. Therefore, I propose a novel instrument to address concerns about the endogenous location of immigrants.

## 4.3 Distance to Historical Exposure Hotspots as Instrument

To capture exogenous variation in the location of Ukrainian workers across Polish counties, I use a novel instrument based on historical records on the forced resettlement of Ukrainians in Poland. In particular, I instrument the current distribution of Ukrainian workers with the distance from historical hotspots of Ukrainian networks that emerged due to Operation Vistula described in section 2.2.

The instrument combines the idea of exploiting historical settlements of immigrants as in a shift-share instrument with an alternatively used instrument based on the distance to the border of the immigrants' origin country. It is inspired by, for example, Dustmann *et al.* (2017) and Aksu *et al.* (2022), which use the distance to the border of the immigrants' origin country as an instrument for the increase in labor supply due to immigrant inflow. Dustmann *et al.* (2017) examine the impact of Czech commuters in German border regions who were daily commuters by law. Aksu *et al.* (2022) study the impact of Syrian refugees in Turkey, where initial registration camps were located close to the Syrian border. While both studies provide credible arguments for using distance to the immigrants' origin country border, they do not apply in the setting of this study. Ukrainian workers were not restricted to commute daily but were allowed to work in Poland for up to 6 months per year and had no other institutional incentives to stay closer to the Ukrainian border.

Therefore, instead of using distance from the Ukrainian border as an instrument, I use distance from the nearest historical forced settlement of Ukrainians. To account for the easier (forced) assimilation of small groups of displaced persons, I focus on larger communities of resettled Ukrainians. First, I geolocate places that were designated as final destinations for forcibly resettled Ukrainians in Operation Vistula. Second, I calculate the relative share of the resettled population by dividing the absolute size of resettlement by the size of the local population in 1950. Third, I define historical hotspots of Ukrainian networks as places where the share of forcibly resettled Ukrainians exceeded 10% of the local population. Then, I calculate the distance in kilometers from the centroid of each county to each of the hotspot points. Finally, I define the instrument as the minimum distance to any of the exposure hotspot points. Figure 5 visualizes the generated instrument.

I test the relevance condition by regressing the share of Ukrainian workers in 2019 against the generated distance instrument. The negative and statistically significant coefficient, presented in Figure 6a, indicates that the distance to the nearest exposure hotspot is a relevant predictor of the contemporary location of Ukrainian workers.<sup>15</sup>

The identifying assumption is the untestable exclusion restriction. In the underlying setting, I assume that the distance from the historical Ukrainian hotspots does not affect changes in local migration patterns beyond the contemporary inflow of temporary workers from Ukraine. As a first check of the plausibility of this assumption, I conduct a balancing test. Similarly to Figure 4, I regress several county characteristics, which potentially have

<sup>&</sup>lt;sup>15</sup>Figure A5 in the Appendix shows a plot of this correlation.



Figure 5: DISTANCE INSTRUMENT

*Notes:* This figure presents the variation in distance (in km) to the nearest exposure hotspot point. Exposure hotspot points are depicted in the map as red points and are defined as localities in which the share of forcibly resettled Ukrainians exceeded 10% of the local population. For visualization purposes, the continuous distance variable is presented with a discrete scale.

an impact on migration patterns, against the generated instrument. Figure 6b shows that none of the estimated coefficients is statistically different from zero. In particular, the instrument is not correlated with pre-treatment changes in local labor market conditions or their initial pre-treatment values.

Furthermore, I regress the instrument on possible historical factors linked to the location of the hotspot points and the changes in migration patterns in the pre-treatment period. The results in Figure 6a reveal that the instrument is not predictive of changes in migration patterns in the pre-treatment period, as the estimated coefficients are never significantly different from zero. The instrument is negatively correlated with factors related to the resettlement of Ukrainians, such as the number of railway stations in 1946, post-World War II destruction, and historical industrial production per capita, that could potentially impact modern-day outcomes.<sup>16</sup> The results are robust to controlling for those factors, thus, taken together these findings support the validity of my instrument.

<sup>&</sup>lt;sup>16</sup>The extent of destruction after World War II is quantified by measuring the volume (in cubic meters) of urban buildings that were destroyed.



(a) FIRST STAGE AND CORRELATES

(b) BALANCING TEST

Figure 6: Correlates of the instrument

Notes: Both panels present bivariate correlations with the instrument (Standardized Beta Coefficients +95% CI). Panel (a): Due to data availability, I use the change in the share of graduates between 2012 and 2013 instead of the change between 2007 and 2013.

## 5 Results and Discussion

## 5.1 Main Results

**OLS Estimates.** I start by providing the results of a linear regression model using Equation 1 in Panel A of Table 1. First, I show results without the inclusion of control variables in Specification 1. Column 1 indicates that there is no evidence of a statistically significant correlation between exposure to the inflow of temporary workers from Ukraine and international out-migration from local labor markets. Next, I include a Bartik (1991) variable, that combines local industry shares at baseline with industry-specific growth rates at the national level to control for labor demand shocks related to industry structure in Specification 2.<sup>17</sup> Then, I include a set of controls for initial conditions such as initial population density, the share of graduates, the share of females, wages, and unemployment in Specification 3.<sup>18</sup> For international out-migration, the coefficient remains statistically not different from zero.

This null result could be a consequence of two factors that drive the coefficient in opposite directions. On the one hand, Ukrainians tend to settle in places with better labor market prospects, which are also likely to be negatively correlated with local out-migration. On the other hand, the complementarity between Ukrainians and Polish emigrants might lead to a positive correlation between the location of Ukrainians and high local international out-migration.

Column 2 shows a consistently negative correlation between exposure to Ukrainian workers and internal out-migration. This suggests that the complementarity effect does not play an important role in internal migration. As expected, due to local amenities, such as better labor market prospects, there is less internal out-migration from places that attract more immigrant workers.

<sup>&</sup>lt;sup>17</sup>Thus, in the setting of this study Bartik variable is defined as  $\Delta Bartik_c = \sum_k (s_{k,c}^{2013} \cdot (\Delta N_k / N_k^{2013}))$ , where k stands for four different sector activities, namely manufacturing and construction, agriculture and trade, financial sector, and other services.

<sup>&</sup>lt;sup>18</sup>Controlling for initial conditions in a first-differences setting allows for post-treatment changes in the relationship between the outcome variable and those county-specific characteristics.

## Table 1: INFLOW OF UKRAINIANS AND LOCAL MIGRATION PATTERNS

	Panel A: OLS				Panel B: 2SLS			
	Out-migr	ation	In-migra	tion	Out-migr	ation	In-migra	tion
Specification	International	Internal	International	Internal	International	Internal	International	Internal
1. Basic regression (no controls)								
Share of Ukrainians	$0.034 \\ (0.066)$	$-0.512^{**}$ (0.194)	$\begin{array}{c} 0.137^{***} \\ (0.033) \end{array}$	$1.162^{***} \\ (0.296)$	$-0.498^{**}$ (0.237)	$-1.886^{***}$ (0.556)	-0.040 (0.125)	-0.719 (1.710)
First-stage F-stat					17.77			
2. Add Bartik variable								
Share of Ukrainians	$0.046 \\ (0.066)$	$-0.508^{**}$ (0.197)	$\begin{array}{c} 0.148^{***} \\ (0.033) \end{array}$	$\begin{array}{c} 1.217^{***} \\ (0.345) \end{array}$	$-0.429^{*}$ (0.228)	$-1.877^{***}$ (0.557)	$0.023 \\ (0.103)$	-0.421 (1.526)
First-stage F-stat						18	.60	
3. Add controls for initial conditions								
Share of Ukrainians	$0.144 \\ (0.096)$	$-0.364^{*}$ (0.190)	$0.085^{*}$ (0.043)	$0.813^{*}$ (0.430)	$-0.529^{**}$ (0.264)	$-1.916^{***}$ (0.583)	-0.073 (0.096)	-0.879 (1.193)
First-stage F-stat						25	.96	
Clusters Observations		7 3'	72 79			7 37	2 79	

Notes: The explanatory variable is the number of statements on the employment of a Ukrainian in 2019 (proxy for the number of Ukrainian workers) standardized by the pre-shock working-age local population in 2013. All dependent variables are standardized by the same denominator as the explanatory variable. I instrument the current distribution of Ukrainian workers using distance from historical hotspots of Ukrainian networks. The controls for initial conditions include population density, the share of females, the share of graduates, the unemployment rate, and average wages in 2013. All regressions are weighted by the pre-treatment working-age population. Standard errors clustered at the NUTS3 level are in parentheses. \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.

Columns 3 and 4 report that counties with stronger exposure to the inflow of temporary workers from Ukraine experience higher internal and international in-migration for a permanent stay. Without the inclusion of control variables, this correlation is highly statistically significant, however, including controls for initial conditions reduces this correlation (see Specification 3). This confirms that the correlation between the inflow of Ukrainian workers and in-migration for permanent stay is driven by the characteristics of the affected counties.

Overall, the findings confirm, that resulting from the endogeneity of immigrant workers' location decisions, the simple OLS regression captures a spurious correlation between the migration patterns of Poles and the presence of Ukrainian workers driven by the (partly unobservable) characteristics of the affected counties. Thus, to interpret the results causally, I resort to the instrumental variables approach.

**2SLS Estimates.** I start again by providing the results without controls in Specification 1 in Panel B of Table 1. Columns 1 and 2 show that, on average, the inflow of Ukrainian workers to local labor markets significantly decreases the international and internal outmigration of the local population. This result is robust to controlling for potential confounders.<sup>19</sup>

The standardization of the outcome and the measure of treatment by the same denominator allows me to interpret the coefficient as the absolute change in the number of emigrants due to the inflow of 100 Ukrainian workers. On average, an inflow of 1000 workers into a county decreases the international out-migration of the local population by around 5 and the internal out-migration by around 19 inhabitants. The estimated effect size indicates that the labor supply shock contributed to the overall decrease in permanent emigration from Poland between 2013 and 2019 by around 30%.<sup>20</sup>

<sup>&</sup>lt;sup>19</sup>Table A2 and Table A3 in the Appendix show that the results remain robust even with the inclusion of several time-varying controls and historical correlates, such as the number of railway stations in 1946, which is highly correlated with the instrument.

<sup>&</sup>lt;sup>20</sup>The total decrease in emigration from Poland between 2013 and 2019 was 21 thousand. The inflow of Ukrainians can be attributed to the overall observed crease in the emigration of around seven thousand inhabitants (calculated as the change in the Ukrainian workers divided by 1000 and multiplied with the point estimate of around 5). Seven thousand divided by 21 thousand equals around 30%.

Columns 3 and 4 show no statistically significant evidence of a decrease in international or internal in-migration to counties with higher exposure to Ukrainian workers. However, while not statistically significant, the size of the negative point estimate for internal in-migration may suggest that the inflow of Ukrainians crowds out in-migration from other Polish counties.

**Pre-trends and dynamic effect.** Figure 7 shows dynamic difference-in-differences coefficients in an event study graph.<sup>21</sup> Although some pretreatment coefficients are statistically significantly different from zero, overall there is no evidence of clear pretreatment trends. This suggests that the parallel trend assumption holds.

The observed effects are in line with the findings discussed earlier with one exception. The dynamic representation reveals statistically significant crowding-out of in-migration from other Polish counties as an initial response to the shock in counties with higher exposure to temporary workers from Ukraine. <sup>22</sup> This effect decreases in magnitude and becomes statistically insignificant for international in-migration after two and for internal in-migration after four years, indicating a reduction of this effect in the long run. However, the crowding-out effect seems to reappear during the COVID-19 pandemic.

While the overall effect on local migration seems to balance out in the short run, in the long run, the magnitude of the decrease in out-migration appears to surpass the effect on in-migration. Furthermore, taking into account the size of the inflow of temporary workers from Ukraine, the results clearly indicate an enlargement of local labor markets.

## 5.2 Validity of the Empirical Results and Robustness Checks

The main identification assumption is that the distance from historical Ukrainian settlements, which were established due to forced resettlement shortly after World War II, has no effect on migration patterns beyond the inflow of temporary workers from Ukraine.

 $<sup>^{21}</sup>$ I control for the number of railway stations in 1946, which is highly correlated with the instrument.

 $<sup>^{22}</sup>$ This finding is in line with recent studies such as Dustmann *et al.* (2017), Amior and Manning (2018), and Monras (2020).



*Notes:* The coefficients to the left of the vertical gray line show the trends before treatment. The coefficients are obtained by interacting the instrumented inflow of Ukrainians with annual indicators. I control for the number of railway stations in 1946. The reported standard errors were obtained using a bootstrap procedure.

23

If this assumption holds, I identify the local average treatment effect (LATE) for the subgroup of compliers to the instrument. In the context of this study, compliers are Ukrainian workers whose location decision was affected by the geographic proximity to the location of their historical ethnic networks. Given that the inflow of Ukrainian workers was unexpected, it is likely that Ukrainian ethnic networks in Poland played a crucial role in the location decision of Ukrainian workers and the group of compliers is sizable. This expectation is reinforced by the strong first stage.<sup>23</sup> The dynamic analysis reveals no indications of pre-treatment trends in migration outcomes between counties with different exposure to the inflow of Ukrainians, which suggests that the parallel trend assumption holds. Furthermore, to test the internal validity of the results, I run several robustness checks.

The Measure of the labor supply shock. I examine whether the results are sensitive to using alternative measures of the labor supply shock. First, in the baseline results, I do not consider seasonal workers as part of the labor supply shock due to the different legal and economic nature of such migration. Panel A in Table A4 in the Appendix shows that the results are robust to the inclusion of seasonal workers in the measure of the labor supply shock. Second, due to the migratory activity distortion during the pandemic, in the baseline results, I do not consider the measure of the labor supply shock from 2020 but from 2019 only. Panel B in Table A4 in the Appendix shows that the results are robust to using the declarations on the employment of a foreigner from 2020.

Furthermore, as both my treatment and outcome variables are standardized by the same denominator, there is a potential risk of spurious correlation. Table A5 in the Appendix shows that the results are robust to the implementation of the Kronmal (1993) specification correction, as suggested by Clemens and Hunt (2019) to address this concern.

Alternative definition of the instrument. I test if the results are robust to several alternative definitions of the instrument. Table A6 in the Appendix shows the results using the alternative thresholds of 12%, 8%, as well as 6% of forcibly resettled Ukrainians in the

<sup>&</sup>lt;sup>23</sup>The first-stage F-test statistics are larger than the conventional lower bounds in all specifications, showing that the instrument is relevant and does not suffer from the weak instrument problem.

local population. The results using several other thresholds than 10% when generating the distance instrument are similar to those reported in the main analysis. Moreover, the results are robust to using a more continuous measure of the instrument. Table A7 in the Appendix shows results using as an alternative instrument the interaction of the historical share of Ukrainians with the distance to the hotspots.

Spatial spillovers and sensitivity to sample restrictions. To assess the robustness of the results to accounting for potential spatial spillovers, I present in Table A8 in the Appendix the results with standard errors following Conley (1999, 2008). The findings remain robust when allowing for spatial correlation within various cutoff distances. Furthermore, a possible source of concern is that the results may be driven by an outlier, such as a large city or a few counties in the commuting zone of a large city. Thus, I conduct a so-called leave-one-out approach. Table A9 and Table A10 in the Appendix provide results where I exclude for estimation in each row one of the six largest functional urban areas (FUA) in Poland.<sup>24</sup> Table A10 provides also two further robustness checks: in Panel C I exclude all counties from the six largest FUAs and in Panel D all 58 urban cores of all FUAs in Poland.

Furthermore, I examine if the results are not driven by counties with high initial levels of migration. In particular, Table A11 in the Appendix shows results where I exclude in each case 5% of counties with the highest initial international out-migration, the highest initial international in-migration, and the highest initial international in-migration, and the highest initial internal in-migration.<sup>25</sup>

McKenzie and Rapoport (2010) find that the self-selection of immigrants may vary with the strength of the migration network. Thus, as my instrument is based on the network of Ukrainians in Poland, the results may be potentially driven by either positively or negatively self-selected migrants. Therefore, as an additional robustness check, in

<sup>&</sup>lt;sup>24</sup>Each of the FUAs consists of counties in a common commuting zone. Figure A6 in the Appendix provides a map with a visualization of the six largest FUAs in Poland.

<sup>&</sup>lt;sup>25</sup>The results are more pronounced when excluding counties with the highest initial internal in-migration. This suggests that counties initially less attractive for internal migration experience a greater decrease in out-migration compared to those that were particularly attractive for internal in-migration prior to the inflow of Ukrainian workers.

Table A12 in the Appendix, I exclude 5% of counties closest to the historical hotspots of Ukrainian networks in Panel A and 5% of counties with the greatest distance to those exposure hotspots in Panel B.

Taken together, the results show that the magnitude and significance of coefficients of interest do not depend on the exclusion of several potential outliers and are not sensitive to the changes in the treatment measure and instrument definition. Moreover, I find no evidence of pre-treatment trends in migration outcomes between counties with different exposure to the inflow of Ukrainians, which suggests that the parallel trend assumption holds.

## 5.3 Mechanisms

The massive and unexpected supply shock of workers with low reservation wage could potentially further increase the out-migration rates, for example, by intensifying competition and putting downward pressure on wages in the affected local labor markets (Borjas, 2003). However, I find that the inflow of Ukrainians leads to a decrease in out-migration. In this section, I analyze the underlying mechanisms.

I first examine the impact of stronger exposure to the inflow of Ukrainian workers on average local wages. Figure 8 shows that there is no evidence of a decrease in wages, on the contrary, although not statistically significant, the point estimate is positive.<sup>26</sup> There are two possible explanations. First, a binding minimum wage could protect native workers who may be in competition with immigrants (Edo and Rapoport, 2019).<sup>27</sup> Second, the inflow of workers from Ukraine does not necessarily lead to increased competition in local labor markets, particularly if immigrants and natives are complements rather than substitutes.<sup>28</sup> Section 2.3 provides descriptive evidence on the complementarity between Ukrainians with low reservation wages and Poles, who often emigrate due to a lack of job

<sup>&</sup>lt;sup>26</sup>Table A13 in the Appendix provides the complete regression results.

<sup>&</sup>lt;sup>27</sup>Poland has a long tradition of minimum wages, which were first introduced as early as 1956. Figure A7 in the appendix shows the ratio of the minimum wage to average wages by county, indicating in which regions the minimum wage might play a particular role in protecting low-income workers.

 $<sup>^{28}</sup>$ See for example Peri and Sparber (2009), Foged and Peri (2016), Mitaritonna *et al.* (2017), Akgündüz and Torun (2020), and Storm (2022).



Figure 8: SCALING-UP OF LOCAL LABOR MARKETS

Notes: This figure presents Standardized Beta Coefficients (+95% CI) using 2SLS. The explanatory variable is the number of statements on the employment of a Ukrainian in 2019 (proxy for the number of Ukrainian workers) standardized by the pre-shock working-age local population in 2013. All dependent variables are standardized by the same denominator as the explanatory variable. I instrument the current distribution of Ukrainian workers using distance to historical hotspots of Ukrainian networks. Controls include population density, the share of females, the share of graduates, the unemployment rate as well as average wages in 2013.

prospects in the Polish labor market saturated with highly educated workers.

The remaining results presented in Figure 8 show evidence of upscaling, that is, an enlargement of local labor markets. In particular, I find evidence of an increase in job offers and employment as well as a decrease in unemployment in counties that are more exposed to the inflow of Ukrainian workers. I also find suggestive evidence of an increase in the number of firms with more than 250 workers. The point estimate on the change in the number of smaller firms is positive as well but estimated very imprecisely. In line with the increase in the number of firms, I find suggestive evidence of higher capital investments in counties more exposed to the inflow of Ukrainians.

Taken together, the results suggest an upscaling, that is, an enlargement of local labor markets, as a potential driver of the decrease in out-migration. On one hand, the sudden inflow of temporary workers from Ukraine with low reservation wages appears to foster the emergence of firms in counties with stronger exposure to this shock, leading to an increase in job offers and overall employment, as well as a decrease in unemployment. On the other hand, descriptive evidence underscores the abundance of highly educated natives at the time of the shock, highlighting the complementarity between Ukrainian workers and Polish emigrants. Consequently, highly educated natives, who could potentially emigrate, may now fill newly created non-manual positions, such as managers, accountants, or supervisors, which complement the jobs performed by Ukrainian workers. Furthermore, firms increase their investment in capital that is complementary to the high-skilled natives, medium-skilled natives, and immigrant workers (Lewis, 2011). This, in turn, contributes to the further enhancement of the local labor markets. Overall, the results indicate that the inflow of Ukrainian workers with low reservation wages facilitated the absorption of abundant natives into the local labor markets, preventing them from emigrating.

## 6 Conclusion

This paper investigates the effect of the unexpected and massive inflow of temporary workers that resulted from Russia's aggression against Ukraine in 2014 on the migration patterns in Poland. The empirical strategy relies on variation across local labor markets in the exposure intensity to Ukrainian workers in a first-difference framework. To address the endogenous location of immigrant workers, I propose a novel instrument based on unique historical data on the forced resettlement of Ukrainians in Poland after World War II.

I find that the inflow of Ukrainians into local labor markets reduces out-migration. On average, the inflow of 1000 Ukrainian workers into a county decreases local internal outmigration by around 19 and international out-migration by approximately 5 inhabitants. Furthermore, I find evidence of a decrease in in-migration as an initial response to the shock; however, this crowding-out effect diminishes after the first few years. The evidence on mechanisms suggests that local labor markets with larger exposure to Ukrainian workers scale up. In particular, I find evidence of an increase in job offers and employment in counties more exposed to Ukrainian workers. Given the suggestive evidence of complementarity between immigrant workers and native emigrants, this may indicate that potential native emigrants are absorbed by the enlarged local labor markets.

The findings have important policy implications, especially in light of the escalation of Russian aggression against Ukraine in 2022 and the subsequent decision to invoke the Temporary Protection Directive in the European Union. The latter has resulted in a significant increase in labor supply not only in Poland but also in other post-communist countries in Eastern and Central Europe, such as Lithuania or Latvia. These countries have also witnessed a notable rise in the number of highly educated citizens since the fall of the Iron Curtain and until recently were also characterized by high emigration rather than immigration. The results show that, in transition countries with an abundance of highly educated citizens, an immigration shock can mitigate the emigration from hosting communities. Thus, this paper points out the potential benefits of open borders and enabling access to the labor market, even for countries in transition without established immigration structures.

## References

- ADSERA, A. and PYTLIKOVA, M. (2015). The Role of Language in Shaping International Migration. The Economic Journal, 125 (586), F49–F81.
- AKGÜNDÜZ, Y. E., ALDAN, A. and BAGIR, Y. K. (2021). Immigration and Inter-Regional Job Mobility: Evidence from Syrian Refugees in Turkey. Working Paper No. 1461. Economic Research Forum (ERF).
- and TORUN, H. (2020). Two and a Half Million Syrian Refugees, Tasks and Capital Intensity. Journal of Development Economics, 145, 102470.
- AKSU, E., ERZAN, R. and KIRDAR, M. G. (2022). The Impact of Mass Migration of Syrians on the Turkish Labor Market. *Labour Economics*, 76, 102183.
- AMIOR, M. and MANNING, A. (2018). The Persistence of Local Joblessness. American Economic Review, 108 (7), 1942–1970.
- BARSBAI, T., RAPOPORT, H., STEINMAYR, A. and TREBESCH, C. (2017). The Effect of Labor Migration on the Diffusion of Democracy: Evidence from a Former Soviet Republic. American Economic Journal: Applied Economics, 9 (3), 36–69.
- BARTIK, T. J. (1991). Who Benefits from State and Local Economic Development Policies? Kalamazoo: WE Upjohn Institute for Employment Research.
- BATUT, C. and SCHNEIDER-STRAWCZYNSKI, S. (2022). Rival Guests or Defiant Hosts? The Local Economic Impact of Hosting Refugees. *Journal of Economic Geography*, 22 (2), 327–350.
- BECKER, S. O., GROSFELD, I., GROSJEAN, P., VOIGTLANDER, N. and ZHURAVSKAYA, E. (2020). Forced Migration and Human Capital: Evidence from Post-WWII Population Transfers. *American Economic Review*, **110** (5), 1430–63.
- BEINE, M. and COULOMBE, S. (2018). Immigration and Internal Mobility in Canada. Journal of Population Economics, 31 (1), 69–106.

- BOHNET, L., PERALTA, S. and PEREIRA DOS SANTOS, J. (2022). Cousins from Overseas: The Labour Market Impact of a Major Forced Return Migration Shock. Discussion Paper No. 15595. IZA Institute of Labor Economics.
- BORJAS, G. J. (2003). The Labor Demand Curve is Downward Sloping: Reexamining the Impact of Immigration on the Labor Market. *The Quarterly Journal of Economics*, 118 (4), 1335–1374.
- (2006). Native Internal Migration and the Labor Market Impact of Immigration. Journal of Human Resources, 41 (2), 221–258.
- BREDTMANN, J., NOWOTNY, K. and OTTEN, S. (2020). Linguistic Distance, Networks and Migrants' Regional Location Choice. *Labour Economics*, **65**, 101863.
- CARD, D. (2001). Immigrant Inflows, Native Outflows, and the Local Labor Market Impacts of Higher Immigration. *Journal of Labor Economics*, **19** (1), 22–64.
- CLEMENS, M. A. and HUNT, J. (2019). The Labor Market Effects of Refugee Waves: Reconciling Conflicting Results. *ILR Review*, **72** (4), 818–857.
- CONLEY, T. G. (1999). GMM Estimation with Cross Sectional Dependence. Journal of Econometrics, **92** (1), 1–45.
- (2008). Spatial Econometrics. in The New Palgrave Dictionary of Economics. London: Palgrave.
- DEVICTOR, X., DO, Q.-T. and LEVCHENKO, A. A. (2021). The Globalization of Refugee Flows. Journal of Development Economics, **150**, 102605.
- DOCQUIER, F., PERI, G. and RUYSSEN, I. (2014). The Cross-country Determinants of Potential and Actual Migration. *International Migration Review*, **48** (1), 37–99.
- DUSTMANN, C., KU, H. and SUROVTSEVA, T. (2021). Real Exchange Rates and the Earnings of Immigrants. Discussion Paper Series CDP 10/21, Centre for Research and Analysis of Migration (CReAM).

- —, SCHÖNBERG, U. and STUHLER, J. (2016). The Impact of Immigration: Why do Studies Reach Such Different Results? *Journal of Economic Perspectives*, **30** (4), 31–56.
- —, and STUHLER, J. (2017). Labor Supply Shocks, Native Wages, and the Adjustment of Local Employment. *The Quarterly Journal of Economics*, **132** (1), 435–483.
- DZIENNIK GAZETA PRAWNA (2012a). Dyplomowani Bezrobotni. January 31, 2012. https: //edgp.gazetaprawna.pl/biblioteka-wydan/2012-01-01 (accessed February 21, 2023).
- DZIENNIK GAZETA PRAWNA (2012b). Padl rekord bezrobocia osob z wyzszym wyksztalceniem. https://serwisy.gazetaprawna.pl/praca-i-kariera/artykuly/613854, padl-rekord-bezrobocia-osob-z-wyzszym-wyksztalceniem.html (accessed February 21, 2023).
- EDO, A. (2019). The Impact of Immigration on the Labor Market. Journal of Economic Surveys, 33 (3), 922–948.
- (2020). The Impact of Immigration on Wage Dynamics: Evidence from the Algerian Independence War. Journal of the European Economic Association, 18 (6), 3210–3260.
- —, RAGOT, L., RAPOPORT, H., SARDOSCHAU, S., STEINMAYR, A. and SWEETMAN, A. (2020). An Introduction to the Economics of Immigration in OECD Countries. *Canadian Journal of Economics*, **53** (4), 1365–1403.
- and RAPOPORT, H. (2019). Minimum Wages and the Labor Market Effects of Immigration. Labour Economics, 61, 101753.
- ELMALLAKH, N. and WAHBA, J. (2023). Syrian Refugees and the Migration Dynamics of Jordanians: Moving In or Moving Out? *Economic Development and Cultural Change*, **71** (4), 1283–1330.
- FINANCIAL TIMES (2014). Exodus of Youth Ages Poland's Population. https://www.ft. com/content/41b93930-52c7-11e4-9221-00144feab7de (accessed February 21, 2023).

- FOGED, M. and PERI, G. (2016). Immigrants' Effect on Native Workers: New Analysis on Longitudinal Data. American Economic Journal: Applied Economics, 8 (2), 1–34.
- GIESING, Y. and SCHIKORA, F. (2023). Emigrants' Missing Votes. European Journal of Political Economy, 78, 102398.
- GIULIETTI, C., WAHBA, J. and ZENOU, Y. (2018). Strong versus weak ties in migration. European Economic Review, **104**, 111–137.
- HAN, J., HUR, J., LEE, J. and YANG, H. (2022). To Move or not to Move? Immigration and Natives' Neighborhood Choices in Seoul, Korea. *Journal of Economic Geography*, 22 (4), 779–799.
- HATTON, T. J. and TANI, M. (2005). Immigration and Inter-regional Mobility in the UK, 1982–2000. The Economic Journal, 115 (507), F342–F358.
- IMF (2014). 25 Years of Transition: Post-Communist Europe and the IMF. Regional Economic Issues Special Report, International Monetary Fund.
- IMF (2016). Emigration and Its Economic Impact on Eastern Europe. IMF Stuff Discussion Note, International Monetary Fund.
- KRONMAL, R. A. (1993). Spurious Correlation and the Fallacy of the Ratio Standard Revisited. Journal of the Royal Statistical Society. Series A (Statistics in Society), 156, 379–392.
- LEWIS, E. (2011). Immigration, Skill Mix, and Capital Skill Complementarity. *The Quarterly Journal of Economics*, **126** (2), 1029–1069.
- MCKENZIE, D. and RAPOPORT, H. (2010). Self-Selection Patterns in Mexico-US Migration: the Role of Migration Networks. *the Review of Economics and Statistics*, **92** (4), 811–821.
- MISIŁO, E. (2013). Akcja "Wisła" 1947: Dokumenty i Materiały. Archiwum Ukraińskie.

- MITARITONNA, C., OREFICE, G. and PERI, G. (2017). Immigrants and firmsâ outcomes: Evidence from france. *European Economic Review*, **96**, 62–82.
- MOCETTI, S. and PORELLO, C. (2010). How Does Immigration Affect Native Internal Mobility? New Evidence from Italy. *Regional Science and Urban Economics*, **40** (6), 427–439.
- MONRAS, J. (2020). Immigration and Wage Dynamics: Evidence from the Mexican Peso Crisis. Journal of Political Economy, 128 (8), 3017–3089.
- MORAGA, J. F.-H., FERRER-I CARBONELL, A. and SAIZ, A. (2019). Immigrant Locations and Native Residential Preferences: Emerging Ghettos or New Communities? *Journal of Urban Economics*, **112**, 133–151.
- MORALES, J. S. (2018). The Impact of Internal Displacement on Destination Communities: Evidence from the Colombian Conflict. Journal of Development Economics, 131, 132– 150.
- NASZ WYBIR (2016). Jak Tworzyla sie Ukrainska Mniejszosc w Polsce?. https://pl. naszwybir.pl/tworzyla-sie-ukrainska-mniejszosc-polsce/ (accessed February 21, 2023).
- OECD (2022a). International Migration Outlook 2022. Organisation for Economic Cooperation and Development.
- OECD (2022b). The Potential Contribution of Ukrainian Refugees to the Labour Force in European Host Countries. Organisation for Economic Co-operation and Development.
- OECD (2023). Population with Tertiary Education (Indicator). Organisation for Economic Co-operation and Development. doi: 10.1787/0b8f90e9-en.
- ONET (2012a). Oszukane Pokolenie. https://wiadomosci.onet.pl/oszukane-pokolenie/ kvxh6 (accessed February 21, 2023).
- ONET (2012b). Spadamy. https://wiadomosci.onet.pl/spadamy/3ehgr (accessed February 21, 2023).

- ORTEGA, J. and VERDUGO, G. (2022). Who Stays and who Leaves? Immigration and the Selection of Natives across Locations. *Journal of Economic Geography*, **22** (2), 221–260.
- PERI, G. (2016). Immigrants, Productivity, and Labor Markets. Journal of Economic Perspectives, **30** (4), 3–30.
- and SPARBER, C. (2009). Task Specialization, Immigration, and Wages. American Economic Journal: Applied Economics, 1 (3), 135–69.
- PLOKHY, S. (2016). The Gates of Europe: A History of Ukraine. Penguin Books.
- RADIO POZNAN (2011). Gdzie po Studiach? Zmywak, Bezrobocie? https://radiopoznan. fm/informacje/pozostale/gdzie-po-studiach-zmywak-bezrobocie (accessed February 21, 2023).
- STATISTICS POLAND (2020). Populacja Cudzoziemcow w Polsce w Czasie COVID-19. Central Statistical Office of Poland.
- STATISTICS UKRAINE (2017). Statistical Bulletin: Ukrainian External Labour Migration. State Statistics Service of Ukraine.
- STORM, E. (2022). Task Specialization and the Native-Foreign Wage Gap. *LABOUR*, **36** (2), 167–195.
- UNHCR (2023). Ukraine Refugee Situation. United Nations High Commissioner for Refugees. https://data.unhcr.org/en/situations/ukraine (accessed February 21, 2023).
- UNITED NATIONS (2022). How is Life: Licro-narratives on the Impact of the Ukraine Crisis in the Republic of Moldova. United Nations Development Programme.
- VERME, P. and SCHUETTLER, K. (2021). The Impact of Forced Displacement on Host Communities: A Review of the Empirical Literature in Economics. *Journal of Development Economics*, **150**, 102606.

## Appendix



**Figure A1:** REFUGEES FROM UKRAINE IN OECD COUNTRIES *Notes:* This figure shows the recorded number of refugees from Ukraine in OECD countries as of mid-September 2022. Data source: OECD (2022a).



(a) ORTHODOX CHURCH IN GOROWO ILAWIECKIE



(b) GREEK CATHOLIC CHURCH IN GODKOWO

#### Figure A2: UKRAINIAN CHURCHES IN POLAND

*Notes:* Examples of churches founded by the descendants of Ukrainians resettled as part of Operation Vistula. Panel (a): Orthodox church of the Dormition of the Holy Mother of God in Gorowo Ilaweckie (Cerkiew prawoslawna pod wezwaniem Zasniecia Najswietszej Bogurodzicy w Gorowie Ilaweckim). Panel (b): Greek Catholic church of the Protection of the Holy Mother of God in Godkowo (Cerkiew Greckokatolicka pod wezwaniem Opieki Przenajswietszej Bogurodzicy w Godkowie). Source: photographs taken by the author.



(a) School Complex in Gorowo Ilaweckie



(b) PRIMARY SCHOOL IN BARTOSZYCE

Figure A3: SCHOOLS WITH UKRAINIAN AS THE LANGUAGE OF TEACHING IN POLAND *Notes:* Examples of schools in northern Poland with Ukrainian as the language of teaching. Panel (a): School Complex with Ukrainian as the Language of Teaching in Gorowo Ilaweckie (Zespol Szkol z Ukrainskim Jezykiem Nauczania w Gorowie Ilaweckim) includes high school and elementary school. Panel (b): Lesya Ukrainka Primary School No. 8 with Ukrainian as the Language of Teaching (Szkola Podstawowa nr 8 im. Lesi Ukrainki z Ukrainskim Jezykiem Nauczania). Source: photographs taken by the author.



**Figure A4:** MIGRATION PATTERNS OF POLES OVER TIME *Notes:* Permanent emigration, permanent immigration, and net migration in Poland over time. The vertical gray lines indicate the period before and after the Russian aggression against Ukraine. Data source: Statistics Poland (2022).



**Figure A5:** SHARE OF UKRAINIANS AND THE DISTANCE INSTRUMENT *Notes:* This figure plots the share of Ukrainian workers against the distance of the centroid of each county to the nearest historical exposure hotspot point. The size of each circle reflects the size of the working-age population in each county in 2013.



Figure A6: FUNCTIONAL URBAN AREAS IN POLAND

*Notes:* This figure presents the six largest Functional Urban Areas (FUAs) in Poland. Each of the FUAs consists of counties in a common commuting zone.



*Notes:* This figure presents the spatial variation in the ratio of minimum wage to average wages. The larger the indicator, the greater the potential role that minimum wages play in the respective counties.

	2013		2019		$\Delta$ 201	3-2019
	Mean	$\operatorname{StD}$	Mean	$\operatorname{StD}$	Mean	$\operatorname{StD}$
International out-migration $^{a}$	0.064	(0.084)	0.022	(0.034)	-0.042	(0.065)
Internal out-migration <sup><math>a</math></sup>	0.579	(0.672)	0.637	(0.714)	0.058	(0.058)
International in-migration <sup><math>a</math></sup>	0.017	(0.034)	0.020	(0.069)	0.003	(0.040)
Internal in-migration <sup><math>a</math></sup>	0.579	(1.163)	0.637	(1.399)	0.058	(0.325)
Average wages <sup><math>a</math></sup>	3.307	(0.506)	4.455	(0.636)	1.147	(0.206)
Unemployment rate	0.099	(0.034)	0.044	(0.022)	-0.055	(0.018)
Employment rate	0.293	(0.119)	0.347	(0.143)	0.053	(0.036)
Job offers <sup><math>a</math></sup>	1.175	(1.238)	1.982	(2.531)	0.807	(2.262)
Firms with $< 250$ workers <sup><i>a</i></sup>	143.2	(44.48)	166.0	(51.58)	22.88	(9.556)
Firms with $> 250$ workers <sup><i>a</i></sup>	0.122	(0.099)	0.126	(0.102)	0.004	(0.026)
Capital stock <sup><math>a</math></sup>	30.02	(27.06)	42.36	(38.64)	12.33	(15.75)
Share of women	0.511	(0.009)	0.511	(0.009)	-0.005	(0.018)
Population density	0.381	(0.675)	0.369	(0.654)	-0.012	(0.085)
Share of graduates	0.007	(0.019)	0.005	(0.014)	-0.003	(0.007)
Share of Ukrainians			0.052	(0.055)		
Distance from hotspot $point^b$			1.770	(1.198)		
Number of railway stations <sup><math>c</math></sup>			13.22	(11.58)		
Destruction after $WWII^c$			1774.7	(7621.8)		
Industrial production per capita <sup><math>d</math></sup>			6.885	(0.528)		
Observations			379			

 Table A1: DESCRIPTIVE STATISTICS

Notes: <sup>a</sup> in 1000, <sup>b</sup> in 100 km, <sup>c</sup> in 1946, <sup>d</sup> in 1954.

	Out-migr	ation	In-migra	tion		
	International	Internal	International	Internal		
Specification	(1)	(2)	(3)	(4)		
1. Basic 2SLS (no controls)						
Share of Ukrainians	-0.498**	-1.886***	-0.0404	-0.719		
	(0.237)	(0.556)	(0.125)	(1.710)		
First-stage F-stat		17.	77			
2. Add time-varying	controls					
Share of Ukrainians	$-0.571^{*}$	-1.998***	-0.057	-1.466		
	(0.307)	(0.525)	(0.103)	(1.243)		
First-stage F-stat		26.	93			
3. Add controls for in	nitial condition.	8				
Share of Ukrainians	-0.529**	-1.916***	-0.073	-0.879		
	(0.264)	(0.583)	(0.0962)	(1.193)		
First-stage F-stat		25.19				
4. Add time-varying	and initial cond	ditions cont	trols			
Share of Ukrainians	$-0.571^{*}$	-1.998***	-0.057	-1.466		
	(0.307)	(0.525)	(0.103)	(1.243)		
First-stage F-stat		26.93				
Clusters		72	2			
Observations		37	'9			

Table A2: INCLUDING TIME-VARYING CONTROLS, 2SLS

Notes: The explanatory variable is the number of statements on the employment of a Ukrainian in 2019 (proxy for the number of Ukrainian workers) standardized by the pre-shock working-age local population in 2013. All dependent variables are standardized by the same denominator as the explanatory variable. I instrument the current distribution of Ukrainian workers using distance from historical hotspots of Ukrainian networks. The time-varying controls include changes between 2013 and 2019 in the population density, the share of females, the unemployment rate, average wages, and the share of graduates. The controls for initial conditions include population density, the share of females, the unemployment rate, average wages, and the share of graduates in 2013. All regressions are weighted by the pre-treatment working-age population. Standard errors clustered at the NUTS3 level are in parentheses. \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.

	Out-migr	ation	In-migration				
Specification	International (1)	Internal (2)	International (3)	Internal (4)			
1. Basic regression (no controls)							
Share of Ukrainians	$-0.498^{**}$ (0.237)	$-1.886^{***}$ (0.556)	-0.0404 $(0.125)$	-0.719 (1.710)			
First-stage F-stat		17.	77				
2. Control for number	er of railway sta	ations in 19	946				
Share of Ukrainians	$-0.635^{**}$ (0.281)	$-2.320^{***}$ (0.571)	-0.253 (0.196)	-1.883 (2.207)			
First-stage F-stat		17.20					
3. Control for destru	actions after We	orld War II					
Share of Ukrainians	$-0.429^{*}$ (0.227)	$-1.803^{***}$ (0.566)	-0.088 (0.134)	-1.160 (2.012)			
First-stage F-stat		15.	01				
4. Control for indust	rial production	per capita	in 1954				
Share of Ukrainians	$-0.429^{*}$ (0.246)	$-2.140^{***}$ (0.578)	$0.006 \\ (0.171)$	-1.286 (2.276)			
First-stage F-stat		15.	39				
Clusters		7	2				
Observations		37	79				

Table A3: CONTROLING FOR HISTORICAL CORRELATES, 2SLS

	Out-migr	ation	In-migration				
	International	Internal	International	Internal			
	(1)	(2)	(3)	(4)			
Panel A: Include seasonal workers							
Share of Ukrainians	-0.516**	-1.955***	-0.042	-0.745			
	(0.258)	(0.586)	(0.130)	(1.768)			
		. ,	. ,	× ,			
First-stage F-stat	15.15						
Panel B: Statements	on the employ	nent of a U	Ikrainian from .	2020			
Share of Ukrainians	-0.503**	-1.905***	-0.041	-0.726			
	(0.238)	(0.561)	(0.126)	(1.715)			
First-stage F-stat		20.	23				
Clusters		7	2				
Observations		37	'9				

Table A4: Alternative Definition of Treatment, 2SLS

	Out-migr	ation	In-migra	tion	
	International	Internal	International	Internal	
	(1)	(2)	(3)	(4)	
Share of Ukrainians	$-0.490^{***}$ (0.164)	$-1.724^{***}$ (0.577)	0.033 (0.096)	-0.067 (1.369)	
First-stage F-stat	16.17				
Clusters	72				
Observations		37	'9		

Table A5:KRONMAL'S CORRECTION, 2SLS

Notes: The explanatory variable is the number of statements on the employment of a Ukrainian in 2019 (proxy for the number of Ukrainian workers). I instrument the current distribution of Ukrainian workers using distance from historical hotspots of Ukrainian networks. Following Kronmal (1993), I control for the reciprocal of the pre-shock working-age local population in 2013. All regressions are weighted by the pre-treatment working-age population. Standard errors clustered at the NUTS3 level are in parentheses. \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.

	Out-migr	ation	In-migra	tion		
	International	Internal	International	Internal		
	(1)	(2)	(3)	(4)		
Panel A: 10% threshold (baseline)						
Share of Ukrainians	-0.498**	-1.886***	-0.040	-0.719		
	(0.237)	(0.556)	(0.125)	(1.710)		
First-stage F-stat		17.	77			
Panel B: 12% thresh	old					
Share of Ukrainians	-0.502**	-1.882***	-0.043	-0.694		
	(0.233)	(0.544)	(0.123)	(1.66)		
First-stage F-stat		18.	65			
Panel C: 6% threshol	ld					
Share of Ukrainians	-0.504**	-1.869***	-0.065	-0.713		
	(0.249)	(0.519)	(0.117)	(1.468)		
First-stage F-stat		21.53				
Panel D: 8% thresho	ld					
Share of Ukrainians	-0.409*	-1.826***	-0.033	-0.683		
	(0.245)	(0.544)	(0.123)	(1.669)		
First-stage F-stat		19.	11			
Clusters		7	2			
Observations		37	'9			

Table A6: ALTERNATIVE DEFINITION OF INSTRUMENTS, 2SLS

Notes: The explanatory variable is the number of statements on the employment of a Ukrainian in 2019 (proxy for the number of Ukrainian workers) standardized by the pre-shock working-age local population in 2013. All dependent variables are standardized by the same denominator as the explanatory variable. I instrument the current distribution of Ukrainian workers using distance from various definitions of historical hotspots of Ukrainian networks using different thresholds, i.e. relative size of the resettled Ukrainian population during Operation Vistula in the local population. The regressions include no controls. All regressions are weighted by the pre-treatment working-age population. Standard errors clustered at the NUTS3 level are in parentheses.\* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.

	Out-migr	ation	In-migra	tion		
	International	Internal	International	Internal		
	(1)	(2)	(3)	(4)		
Panel A: 10% threshold, instrument interacted with historical share						
Share of Ukrainians	-0.498**	-1.887***	-0.041	-0.721		
	(0.237)	(0.557)	(0.125)	(1.711)		
First-stage F-stat		17.	.72			
Panel B: 12% thresh	old, instrument	interacted	with historical	share		
Share of Ukrainians	-0.503**	-1.884***	-0.043	-0.695		
	(0.233)	(0.544)	(0.123)	(1.661)		
First-stage F-stat		18.60				
Panel C: 6% threshol	ld, instrument a	interacted u	with historical s	hare		
Share of Ukrainians	-0.505**	-1.871***	-0.065	-0.715		
	(0.250)	(0.520)	(0.118)	(1.469)		
First-stage F-stat		21.	.46			
Panel D: 8% thresho	ld, instrument a	interacted u	with historical s	hare		
Share of Ukrainians	-0.410*	-1.827***	-0.033	-0.685		
	(0.246)	(0.544)	(0.123)	(1.670)		
First-stage F-stat		19.05				
Clusters		7	2			
Observations		37	79			

Table A7: Alternative definition of instrument (cont'd), 2SLS

Notes: The explanatory variable is the number of statements on the employment of a Ukrainian in 2019 (proxy for the number of Ukrainian workers) standardized by the pre-shock working-age local population in 2013. All dependent variables are standardized by the same denominator as the explanatory variable. I instrument the current distribution of Ukrainian workers using distance from various definitions of historical hotspots of Ukrainian networks interacted with their historical share using different thresholds, i.e. relative size of the resettled Ukrainian population during Operation Vistula in the local population. The regressions include no controls. All regressions are weighted by the pre-treatment working-age population. Standard errors clustered at the NUTS3 level are in parentheses.\* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.

	Out-migr	ation	In-migration		
	International	Internal	International	Internal	
	(1)	(2)	(3)	(4)	
Panel A: 25 km					
Share of Ukrainians	-0.498**	-1.886***	-0.0404	-0.719	
	(0.212)	(0.557)	(0.125)	(1.808)	
Panel B: 50 km					
Share of Ukrainians	-0.498*	-1.886***	-0.040	-0.719	
	(0.282)	(0.685)	(0.133)	(1.672)	
Panel C: 100 km					
Share of Ukrainians	-0.498*	-1.886***	-0.040	-0.719	
	(0.282)	(0.517)	(0.077)	(NA)	
Panel D: 150 km					
Share of Ukrainians	-0.498**	-1.886***	-0.040	-0.719	
	(0.244)	(0.405)	(0.126)	(1.203)	
Clusters		7	2		
Observations		37	'9		

#### Table A8: CONLEY STANDARD ERRORS, 2SLS

	Out-migr	ation	In-migra	tion
	International	Internal	International	Internal
	(1)	(2)	(3)	(4)
Panel A: Exclude FU	VA Warsaw			
Share of Ukrainians	-0.512**	-1.879***	-0.043	-0.740
	(0.235)	(0.571)	(0.121)	(1.717)
First-stage F-stat		18.	58	
Observations		35	58	
Panel B: Exclude FU	VA Lodz			
Share of Ukrainians	-0.481**	-1.881***	-0.039	-0.719
	(0.229)	(0.549)	(0.123)	(1.686)
First-stage F-stat		19.	30	
Observations		37	'1	
Panel C: Exclude FU	VA Krakow			
Share of Ukrainians	-0.462**	-1.715***	0.049	0.636
	(0.229)	(0.537)	(0.083)	(1.071)
First-stage F-stat		19.	29	
Observations		36	59	
Panel D: Exclude FU	VA Wroclaw			
Share of Ukrainians	-0.597**	-2.458***	-0.089	-1.555
	(0.301)	(0.540)	(0.159)	(2.116)
First-stage F-stat		18.	54	
Observations		37	2	

Table A9: LEAVE-ONE-OUT:	EXCLUDE T	THE LARGEST	FUAS,	2SLS
--------------------------	-----------	-------------	-------	------

	Out-migr	ation	In-migra	tion	
	International	Internal	International	Internal	
	(1)	(2)	(3)	(4)	
Panel A: Exclude FUA Poznan					
Share of Ukrainians	-0.534**	-1.873***	-0.056	-0.784	
	(0.246)	(0.570)	(0.130)	(1.761)	
First-stage F-stat		16.	74		
Observations		37	71		
Panel B: Exclude FU	UA Tricity (Gda	ansk, Gdyn	ia, Sopot)		
Share of Ukrainians	-0.469*	-1.832***	-0.030	-1.237	
	(0.249)	(0.581)	(0.128)	(1.765)	
First-stage F-stat		15.	49		
Observations		37	71		
Panel C: Exclude the	e six largest FU.	As			
Share of Ukrainians	$-0.612^{*}$	-2.253***	-0.007	-0.834	
	(0.318)	(0.601)	(0.105)	(1.296)	
First-stage F-stat		24.	33		
Observations		31	7		
Panel D: Exclude ur	ban cores of all	58 FUAs i	n Poland		
Share of Ukrainians	$-0.515^{*}$	-2.099***	0.065	0.835	
	(0.269)	(0.621)	(0.062)	(0.679)	
First-stage F-stat		13.	68		
Observations		33	34		

	Out-migr	ation	In-migration				
	International (1)	Internal (2)	International (3)	Internal (4)			
Panel A: Exclude co	enties with high initial international out-migr						
Share of Ukrainians	$-0.482^{**}$ (0.224)	$-1.887^{***}$ (0.552)	-0.0380 (0.126)	-0.705 (1.726)			
First-stage F-stat Observations	$\begin{array}{c} 17.26\\ 361 \end{array}$						
Panel B: Exclude counties with high initial internal out-migration							
Share of Ukrainians	$-0.495^{**}$ (0.245)	$\begin{array}{cccc} -0.495^{**} & -1.760^{***} & -0.0 \\ (0.245) & (0.568) & (0.125) \end{array}$		-0.885 (1.797)			
First-stage F-stat Observations	$15.93 \\ 361$						
Panel C: Exclude counties with high initial international in-migration							
Share of Ukrainians	$-0.469^{*}$ (0.256)	$-2.109^{***}$ (0.641)	-0.102 (0.153)	-1.273 (2.063)			
First-stage F-stat Observations	$\begin{array}{c}13.46\\361\end{array}$						
Panel D: Exclude counties with high initial internal in-migration							
Share of Ukrainians	$-0.625^{**}$ (0.262)	$-2.163^{***}$ (0.574)	-0.081 (0.141)	-1.211 (1.972)			
First-stage F-stat Observations	21.13 361						

Table A11: EXCLUDE COUNTIES WITH HIGH INITIAL MIGRATION, 2SLS

	Out-migra	ation	In-migration					
	International	Internal	International	Internal				
	(1)	(2)	(3)	(4)				
Panel A: Exclude counties closest to the hostpots								
Share of Ukrainians	-0.385*	-1.786***	-0.004	-0.474				
	(0.217)	(0.515)	(0.110)	(1.562)				
First-stage F-stat	21.82							
Panel B: Exclude counties with the lowest exposure to hotspots								
Share of Ukrainians	-0.550*	-1.434**	-0.081	-0.341				
	(0.311)	(0.618)	(0.183)	(2.183)				
First-stage F-stat	10.87							
Clusters		72	2					
Observations	379							

## Table A12: SENSITIVITY TO HOTSPOT EXPOSURE, 2SLS

	Average wage (1)	Unemployment rate (2)	Employment rate (3)	Job offers (4)	Firms with $<250$ workers $(5)$	Firms with $>250$ workers (6)	Capital stock (7)
Panel A: Basic 2SLS	G (no contr	ols)					
Share of Ukrainians	$\begin{array}{c} 0.192 \\ (0.259) \end{array}$	-0.335 (0.357)	0.444 (0.366)	$0.298^{**}$ (0.129)	$0.461 \\ (0.409)$	$0.272^{*}$ (0.142)	$\begin{array}{c} 0.435^{***} \\ (0.156) \end{array}$
First-stage F-stat				17.77			
Panel B: Add time-v	arying con	trols					
Share of Ukrainians	$\begin{array}{c} 0.178 \\ (0.279) \end{array}$	-0.357 (0.329)	$0.604^{**}$ (0.243)	$\begin{array}{c} 0.291^{**} \\ (0.139) \end{array}$	$0.599^{**}$ (0.298)	$0.305^{**}$ (0.152)	$0.500^{***}$ (0.178)
First-stage F-stat				20.34			
Panel C: Add control	ls for initid	al conditions					
Share of Ukrainians	$\begin{array}{c} 0.369 \\ (0.269) \end{array}$	$-0.495^{***}$ (0.167)	$0.406^{**}$ (0.172)	$0.326^{**}$ (0.147)	$0.229 \\ (0.263)$	$0.281^{*}$ (0.169)	$\begin{array}{c} 0.241 \\ (0.172) \end{array}$
First-stage F-stat				25.19			
Panel D: Add time-v	arying and	l initial conditions	controls				
Share of Ukrainians	$\begin{array}{c} 0.380 \\ (0.282) \end{array}$	$-0.522^{***}$ (0.182)	$\begin{array}{c} 0.487^{***} \\ (0.170) \end{array}$	$\begin{array}{c} 0.325^{**} \\ (0.150) \end{array}$	$\begin{array}{c} 0.343 \ (0.234) \end{array}$	$0.293^{*}$ (0.171)	$0.259 \\ (0.176)$
First-stage F-stat				26.82			
Clusters				72			

#### Table A13: EVIDENCE OF SCALING-UP OF LOCAL LABOR MARKETS, 2SLS

*Notes:* This table presents Standardized Beta Coefficients using 2SLS. The explanatory variable is the number of statements on the employment of a Ukrainian in 2019 (proxy for the number of Ukrainian workers) standardized by the pre-shock working-age local population in 2013. All dependent variables are standardized by the same denominator as the explanatory variable. I instrument the current distribution of Ukrainian workers using distance from historical hotspots of Ukrainian networks. The time-varying controls include changes between 2013 and 2019 in the population density, the share of females, and the share of graduates. The controls for initial conditions include population density, the share of graduates, the unemployment rate as well as average wages in 2013. All regressions are weighted by the pre-treatment working-age population. Standard errors clustered at the NUTS3 level are in parentheses. \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.